



# **Review Review of Modeling, Modulation, and Control Strategies for the Dual-Active-Bridge DC/DC Converter**

Jiayang He<sup>1</sup>, Yangyu Chen<sup>1</sup>, Jiongtao Lin<sup>1</sup>, Jianghui Chen<sup>2,\*</sup>, Li Cheng<sup>2</sup> and Yu Wang<sup>1,\*</sup>

- <sup>1</sup> School of Automation, Guangdong University of Technology, Guangzhou 510006, China; 2112204533@mail2.gdut.edu.cn (J.H.); 2112304018@mail2.gdut.edu.cn (Y.C.); 2112304478@mail2.gdut.edu.cn (J.L.)
- <sup>2</sup> College of Automation, Zhongkai University of Agriculture and Engineering, Guangzhou 510225, China; chengli@zhku.edu.cn
- \* Correspondence: chenjianghui@zhku.edu.cn (J.C.); yuwang@gdut.edu.cn (Y.W.)

Abstract: This paper provides a comprehensive review of the existing research on the Dual Active Bridge (DAB) DC-DC converter, focusing on modeling methods, modulation strategies, optimization algorithms, and control methods. A comparative analysis of selected methods along with guidelines to assist engineers and researchers in their study of DAB is also presented. Firstly, a comprehensive review of modulation strategies for DAB is provided, ranging from classical phase-shift modulation to the popular asymmetric duty modulation. The intrinsic relationships among different modulation methods are summarized, and a comparison is made based on the difficulty of control and DAB operating characteristics. Secondly, the various modeling methods for DAB are described, including reduced-order modeling, generalized state-space averaging modeling, and discrete-time modeling methods. A comparison is made based on the suitability for different application scenarios, providing recommendations for the adoption of different modeling methods. Furthermore, a survey of optimization algorithms for modulation methods is presented, including classical algorithms, swarm intelligence optimization, and reinforcement learning algorithms. A number of criteria are proposed for different algorithms, and an analysis of the unresolved challenges and future prospects is provided. Finally, the advanced control methods for DAB are summarized based on control effectiveness and applicability. The article concludes with a summary and an outlook on future research directions is also provided.

**Keywords:** dual active bridge (DAB); modeling methods; modulation strategies; optimization algorithms; advanced control methods

# 1. Introduction

With the escalation of global energy crises and environmental issues, the global energy consumption pattern is gradually shifting from traditional fossil fuels towards distributed energy sources such as photovoltaics, wind power, and fuel cells [1]. This trend directly promotes the development of renewable energy utilization equipment (REUE) and energy storage systems (ESSs) [2–4]. Traditional AC distribution networks are gradually revealing their limitations in terms of accommodating new energy sources and ensuring power quality [5]. In comparison to traditional AC systems, DC grids offer independent and rapid control of active and reactive power, enhance power transmission capability and system stability, reduce intermediate stages for energy storage and new energy source integration, lower costs, and improve power conversion efficiency and power quality [6,7]. In a DC power grid, the DC-DC converters require both bidirectional power flow capability and electrical isolation for safety assurance [8]. Currently, this is achieved by utilizing the power electronic converters for high-frequency switching and high-frequency isolation transformers for voltage transformation and electrical isolation [9–12].



Citation: He, J.; Chen, Y.; Lin, J.; Chen, J.; Cheng, L.; Wang, Y. Review of Modeling, Modulation, and Control Strategies for the Dual-Active-Bridge DC/DC Converter. *Energies* **2023**, *16*, 6646. https://doi.org/10.3390/en16186646

Academic Editors: Alon Kuperman and Vítor Monteiro

Received: 19 August 2023 Revised: 6 September 2023 Accepted: 13 September 2023 Published: 15 September 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The DAB was initially proposed by de Doncker et al. [13]. It is considered to be the core circuit for medium to high voltage and high-capacity DC-DC converters, owing to its advantages of bidirectional power flow capability, ease of implementing soft switching, modularity, fast dynamic response, and high power density [14–17]. The DAB has been widely studied and applied as an interface between energy storage devices and DC grids [18,19], as well as in solid-state transformers (SSTs) and charging equipment for new energy vehicles [20–25], as illustrated in Figure 1.



Figure 1. Application of DAB in DC grid.

The basic topology of the DAB is depicted in Figure 2. As can be observed from the figure, there are four power switches on each side of the high-frequency transformer, and the transformer provides isolation between the primary and secondary sides. The leakage inductance is represented by  $L_s$ . By controlling the magnitude and direction of the phase shift between the four bridge arms, energy can freely flow between the two DC power sources [26–29].



Figure 2. The topology of DAB.

In all applications of the DAB, it is essential to first model the DAB converter and design the controller based on the specified steady-state and dynamic performance criteria according to the adopted modulation method. The objective of this paper is to provide a comprehensive review and comparison of different modeling methods, modulation methods, optimization algorithms, and control strategies for DAB converters. Furthermore,

it aims to offer promising research directions for researchers who are newly engaged in DAB studies.

The remaining sections of this paper are organized as follows. Section 2 introduces and compares the classical modulation method for DAB, namely, phase-shift modulation, and the recently proposed modulation method, asymmetric duty modulation. Section 3 presents an overview of the current modeling methods for DAB, including reduced-order modeling, generalized state-space averaging modeling, discrete-time modeling, and general modeling based on the Fourier series, followed by a comparative analysis of these modeling methods. Section 4 provides a comprehensive review and comparison of various optimization approaches and algorithms employed to address optimization problems in DAB, including the KKT approach, genetic algorithms, particle swarm algorithms, and Q-learning algorithms. Section 5 summarizes and compares advanced control algorithms for DAB, including genetic voltage feedback control, model predictive control, model reference adaptive control, and active disturbance rejection control. Finally, the paper concludes with a summary and outlook on future research directions in DAB.

# 2. Modulations of DAB

To enhance the efficiency of DAB operation, the development of modulation strategies has progressed from single-degree-of-freedom approaches to multi-degree-of-freedom strategies, which in turn has facilitated advancements in DAB control methods. During the initial stages of research, the focus of DAB modulation strategy studies primarily centered on phase-shift modulation. A large number of scholars extensively investigated various phase-shift modulation (PSM) strategies in terms of converter circulating power, RMS current, soft-switching range, and power transmission range. In recent years, with the introduction of asymmetric duty modulation (ADM), scholars have continued to analyze the operational characteristics of DAB under this modulation scheme and further enhanced the DAB efficiency by combining it with PWM. Therefore, this section provides a comprehensive review of the various modulation strategies for DAB discussed in the existing literature.

#### 2.1. Phase-Shift Modulation Strategy

The PSM is the most widely used modulation method in the context of dual active bridge (DAB) converters. It can be classified into several categories, including singlephase shift (SPS), extended-phase shift (EPS), dual-phase shift (DPS), and triple-phase shift (TPS). SPS modulation is the most fundamental and widely adopted method in DAB converters [13,30]. However, its applicability is limited in the case of non-matching inputoutput voltage ratios and transformer turns ratio, as it is difficult to achieve zero-voltage switching (ZVS) and can result in significant circulating power [31–33]. Subsequently, EPS and DPS modulation methods were proposed by researchers to overcome these limitations and improve the converter's performance in terms of circulating power, current stress, and ZVS range [34–36]. The TPS modulation method, proposed in [37,38], represents a generalized version of all the PSM methods, and other PSM methods can be considered as special cases of TPS. Although TPS modulation introduces more operating modes and complexity, it offers several advantages, including minimum current stress, minimum power loss, larger ZVS range, and higher power factor, which make it suitable for various applications [39–41]. As all modulation methods can be viewed as special cases of TPS, the TPS modulation waveform is presented in Figure 3, without individual descriptions of other methods.

Assuming  $D = t_p/T_s$  is defined as the phase-shift ratio, where  $t_p$  represents the phase-shift time and  $T_s$  denotes the half-period duration.  $D_1$  represents the internal phase-shift ratio on the primary side of the DAB,  $D_2$  represents the phase-shift ratio between the primary and secondary sides, also known as the external phase-shift ratio, and  $D_3$  represents the internal phase-shift ratio on the secondary side. When  $D_1 = 0$  and  $D_2 = D_3$ , it corresponds to the single-phase-shift modulation. When either  $D_1 = 0$  or  $D_2 = D_3$ , it



corresponds to the extended-phase-shift modulation. When  $D_1 = D_2 - D_3$ , it corresponds to the dual-phase-shift (DPS) modulation.

Figure 3. The working waveform under PSM.

#### 2.2. Asymmetric Duty Modulation Strategy

The asymmetric duty modulation (ADM) strategy is an extension of the phase-shift modulation strategy. From Figure 3, it can be observed that all eight power switches in the DAB remain conducting for half a switching period, and in the phase-shift modulation strategy, the switches within the same bridge arm have the same conduction time. The asymmetric duty modulation strategy allows for different conduction times for switches within the same bridge arm, thereby adding three additional control degrees of freedom. The typical waveform of the ADM is shown in Figure 4.



Figure 4. The working waveform under ADM.

In the ADM strategy, the switches on one bridge arm are not triggered in a complementary fashion with a 50% duty cycle. Specifically, for the primary side, the conduction times of switches  $S_1$  and  $S_2$  are equal, while the conduction times of switches  $S_3$  and  $S_4$ are equal. The conduction times of  $S_3$  and  $S_4$  are less than half a period and are denoted as  $D_1T_s$ , whereas the conduction times of  $S_1$  and  $S_2$  are greater than half a period. Similarly, for the secondary side, the conduction times of switches  $Q_1$  and  $Q_2$  are equal, and the conduction times of switches  $Q_3$  and  $Q_4$  are equal. The conduction times of  $Q_3$  and  $Q_4$  are less than half a period and are denoted as D3, while the conduction times of  $Q_1$  and  $Q_2$  are greater than half a period. Additionally, there is a phase shift  $D_2T_s$  between the conduction times of  $S_1$  and  $Q_1$ . It can be observed that when  $D_1 = D_3 = 0.5$ , ADM essentially becomes PSM. In other words, PSM is a special case of the ADM strategy. The contrast between ADM and PSM resides in the steady-state waveforms produced by PSM, which exhibit symmetry within half a period, resulting in high-frequency AC voltages on both the primary and secondary sides of the transformer containing two zero-voltage segments within one switching cycle. In contrast, the high-frequency AC voltage in ADM displays asymmetry within half a period, containing only one zero-voltage segment within each switching cycle.

The ADM strategy was initially proposed in [42], but its application in efficiency optimization of the DAB was not studied. In [43], ADM was only introduced on the primary side, and the ZVS conditions under four operating modes were analyzed. The authors of [44] proposed a method utilizing the ADM to reduce the peak inductor current. The optimization objective in this scheme is chosen as the peak current, which can be considered as an indicator of the RMS current to some extent. Due to the simplicity of the peak current model and the absence of variable coupling, the computational complexity is greatly reduced. Under light load conditions, the asymmetric pulse width modulation exhibits a smaller RMS current compared with PSM, while under medium and heavy load conditions, the RMS current is larger than that of PSM. To address the problem of narrow soft-switching range and low efficiency of the PSM method over a wide voltage range and under light load conditions, the authors of [45] proposed a unilateral asymmetric duty ratio modulation. The unilateral asymmetric modulation strategy combines the advantages of PSM in terms of Ohmic losses and ADM in terms of soft-switching range. ADM is employed on one side while PSM is used on the other side, achieving comprehensive performance optimization under light-load and voltage mismatch conditions.

#### 2.3. Comparison of Modulation Strategy

Different modulation strategies need to be compared in terms of control difficulty and converter operating characteristics. It is desirable to achieve satisfactory operating characteristics with relatively simple controllers. As the degree of freedom in DAB control increases, the controller design becomes more complex, but the increased flexibility in control degrees allows for more satisfactory DAB operating characteristics. Therefore, the selection of an appropriate modulation method should consider the design conditions. PSM, as it can be classified under the TPS framework, exhibits the characteristic of adjusting phase shift without adjusting the conduction duty cycle of the switches, resulting in waveforms with half-cycle symmetry. PSM has a maximum of three control degrees of freedom, making the controller design relatively simple. However, due to the symmetric nature of the TPS framework, zero-voltage switching (ZVS) operation is lost when there is poor voltage match between the DC bus and the energy storage system, leading to increased switching losses. Furthermore, the inclusion of a substantial quantity of reactive power components raises the RMS of the inductor current, leading to increased conduction losses. Consequently, this decrease in conversion efficiency becomes particularly pronounced when operating under light-load conditions. ADM itself possesses three degrees of freedom, and when combined with phase-shift modulation, it can achieve up to six degrees of freedom. Due to its asymmetric control nature, the controller design becomes more complex. However, ADM can achieve lower RMS current compared with PSM under the same power transmission, wider soft-switching range under voltage mismatch conditions, and higher efficiency under light-load conditions. Moreover, ADM also faces inherent challenges such as complex current stress or RMS current mathematical modeling and variable coupling, making the analytical solution for the optimal solution difficult and online calculations challenging. Additionally, due to the presence of even harmonic components in voltage and current under the ADM scheme, the calculations and optimization results based on the fundamental component analysis model may be inaccurate.

The simulation analysis of output power flexibility for different modulation methods is depicted in Figure 5. The vertical axis represents the transmitted power, while the horizontal axis signifies the external phase-shift ratio value, denoted as  $D_2$ . For a given transmitted power, SPS modulation corresponds to two specific phase-shift ratio values,  $D_2$ . The blue region and the purple region, overlapping with both blue and red, represent the power characteristic range under DPS modulation. In comparison to SPS modulation, DPS modulation exhibits an infinite number of combinations of phase-shift ratio values, denoted as  $D_1$  and  $D_2$ , for the same transmitted power. This significantly augments the power regulation flexibility of the DAB, playing a crucial role in the coordinated optimization of various characteristics. The red and purple regions represent the power range under EPS modulation, which also boasts a high degree of flexibility. The green region signifies the power regulation range under the ADM modulation. It is evident that the ADM modulation not only enhances regulation flexibility but also extends the power range, rendering it an ideal modulation approach.



Figure 5. Power characteristics diagram under different modulation methods.

#### 3. Modeling of DAB

The modeling plays a significant role in analyzing power electronic circuits. On one hand, the establishment of converter models enables a clear analysis of the interrelationships among various physical quantities. On the other hand, when applying classical control theory to analyze nonlinear systems like power electronic circuits, it is typically necessary to first develop a large-signal model and then linearize it to establish a small-signal model for further analysis. Currently, there are four main modeling methods for DAB converters. Among them, the reduced-order modeling method is widely applied in DAB-related research due to its simplicity and clear representation of the DAB converter's operation process. The remaining three methods, namely, state-space averaging, general-ized state-space averaging, and discrete-time modeling, have also found applications in the design of DAB controllers.

# 3.1. Reduced-Order Modeling

By neglecting the dynamic variations in transformer currents and considering the transformer as an ideal component, with the assumption of sufficiently large capacitors on



Figure 6. The reduced-order equivalent circuit of DAB.

Subsequently, the basic operating modes of the DAB were analyzed, and the models for each operating mode were established in the time domain. Based on the boundary conditions at the switching instants, the overall average value model was constructed, and the small-signal model was developed to obtain the system transfer function. The transfer function of the DAB is a first-order model, which reflects the steady-state behavior of the DAB while ignoring its dynamic characteristics. The following is the reduced-order model of DAB using the SPS modulation method.

Introducing the switch state function:

$$s_{1}(\tau) = \begin{cases} 1, & 0 \le \tau < \frac{T}{2} \\ -1, & \frac{T}{2} \le \tau < T \end{cases}$$
(1)

$$s_{2}(\tau) = \begin{cases} 1, & \frac{dT}{2} \le \tau < \frac{T}{2} + \frac{dT}{2} \\ -1, & 0 \le \tau < \frac{dT}{2} \text{ or } \frac{T}{2} + \frac{dT}{2} \le \tau < T \end{cases}$$
(2)

When the power switches  $S_1$  and  $S_4$  are turned on,  $s_1(\tau) = 1$ ; when the power switches  $S_2$  and  $S_3$  are turned on,  $s_1(\tau) = -1$ . When the power switches  $Q_1$  and  $Q_4$  are turned on,  $s_2(\tau) = 1$ , and when  $Q_2$  and  $Q_3$  are turned on,  $s_2(\tau) = -1$ . Due to the dependence of the inductor current polarity on the switching state of the power switches, the states are classified into six categories in the analysis. However, since some circuit state variables remain unchanged between certain stages, the calculation of the mathematical model for the steady-state power of the DAB converter can be divided into four stages. Then, a unified model for the inductor current can be established:

$$\int_{t_{i}}^{t} L_{s} \frac{di_{s}}{dt} dt = \int_{t_{i}}^{t} s_{1}(t) V_{\text{in}} - s_{2}(t) n V_{\text{out}} dt$$
(3)

Based on Equation (3), the  $L_s$  current in each of the four operation modes can be obtained. By utilizing the DAB power definition equation within a single switching period, the DAB power transmission expression can be derived under SPS modulation:

$$P = \frac{1}{T} \int_0^{T_t} v_{AB} i_L dt = \frac{n V_{in} V_{out}}{2f L_s} D(1 - D)$$
(4)

The expressions for the average currents on each side can then be obtained as follows:

$$\begin{cases}
I_{\rm in} = \frac{nV_{\rm out}}{2fL_s}D(1-D) \\
I_{\rm out} = \frac{nV_{\rm in}}{2fL_s}D(1-D)
\end{cases}$$
(5)

Building on the steady-state model described above, the following expressions are obtained by adding the small perturbation signals  $\hat{v}_{in}$ ,  $\hat{v}_{out}$ ,  $\hat{i}_{in}$ , and  $\hat{i}_{out}$  while ignoring higher-order terms:

$$\begin{cases} \hat{i}_{in} = b_1 \hat{d} + b_2 \hat{u}_{out} \\ \hat{i}_{out} = b_3 \hat{d} + b_2 \hat{u}_{in} \end{cases}$$
(6)

Based on Equation (6)  $b_1 = nV_{out}(1-2D)/2fL_s$   $b_2 = nD(1-D)/2fL_s$  and  $b_3 = nV_{in}(1-2D)/2fL_s$ , the small signal model of DAB is obtained, and the equivalent circuit diagram is shown in Figure 7.



Figure 7. The small signal equivalent circuit of DAB.

Based on the equivalent circuit diagram, the transfer functions between variables can be obtained:

$$\begin{cases} G_{\hat{v}_{in}d} = \frac{v_{in}}{\hat{d}} \Big|_{\hat{v}_{out}=0} = \frac{-b_1}{C_1 s} \\ G_{\hat{v}_{in}\hat{v}_{out}} = \frac{\hat{v}_{out}}{\hat{v}_{in}} \Big|_{d=0} = \frac{-b_2 R_2}{R_2 C_2 s + 1} \\ G_{\hat{v}_{out}d} = \frac{\hat{v}_{out}}{\hat{d}} \Big|_{\hat{v}_{in}=0} = \frac{-b_3 R_2}{R_2 C_2 s + 1} \end{cases}$$
(7)

# 3.2. Generalized State-Space Averaging Model

The generalized state-space averaging method was proposed to address the limitations of the traditional state-space averaging method in describing the variations in AC quantities [49]. By considering higher-order Fourier coefficients of the state variables, the generalized state-space method can closely approximate the actual waveforms, capturing significant ripple components and higher-frequency oscillations. In [50], the generalized state-space averaging modeling approach was first applied to DC-DC converters. The traditional state-space averaging method neglects high-frequency fluctuations in power electronic circuits. In the case of the DAB converter, the inductor current exhibits significant variations. Applying the state-space averaging method requires a reduction in the model order, which compromises accuracy and fails to capture the dynamic response of the inductor current. Therefore, in [51], the generalized state-space averaging modeling method was employed for the DAB converter. Based on the representation of waveforms using complex Fourier series, the periodic variable x(t) can be expressed as a Fourier series, and the generalized averaging method is applied:

$$x(t) = \sum_{k=-\infty}^{\infty} \langle x \rangle_k(t) e^{jk\omega t}$$
(8)

In the equation,  $\omega$  represents the fundamental angular frequency, and  $\langle x \rangle_k(t)$  is the sliding average coefficient obtained by moving a time window with a width of *T* along the time axis. Its mathematical definition is:

$$\langle x \rangle_k(t) = \frac{1}{T} \int_{t-T}^t x(\tau) e^{-jk\omega\tau} d\tau.$$
(9)

Two fundamental properties can be derived from Equations (8) and (9), as shown below:

$$\frac{d}{dt}\langle x\rangle_k(t) = \left\langle \frac{d}{dt}x \right\rangle_k(t) - jk\omega\langle x\rangle_k(t)$$
(10)

This property signifies that the derivative of a function's Fourier series coefficients can be expressed by first differentiating the function and then taking its Fourier series coefficients. As a result, the Fourier series coefficient equation corresponding to the state variable equation can be established.

$$\langle xy \rangle_k = \sum_{i=-\infty}^{\infty} \langle x \rangle_{k-i} \langle y \rangle_i \tag{11}$$

Another property of the Fourier transform is its frequency domain convolution property, which is mainly used in power electronics converter applications to convert the Fourier series coefficients of the product of switch variables and state variables into the product of their respective Fourier series coefficients.

For systems with insignificant ripple, the fundamental and DC components of the Fourier series coefficients can accurately reflect the variations in the state variables. Therefore, it is sufficient to consider k = 0,  $k = \pm 1$ . Since the Fourier coefficients for k = 1 and k = -1 are complex conjugates, the real and imaginary parts of the k = 1 coefficient can be used to represent the Fourier coefficients for  $k = \pm 1$ . By incorporating Equation (11), the Fourier coefficient equations for the system can be derived:

$$\begin{cases}
\langle xy \rangle_{0} = \langle x \rangle_{0} \langle y \rangle_{0} + 2 \left( \langle x \rangle_{1}^{R} \langle y \rangle_{1}^{R} + \langle x \rangle_{1}^{I} \langle y \rangle_{1}^{I} \right) \\
\langle xy \rangle_{1}^{R} = \langle x \rangle_{0} \langle y \rangle_{1}^{R} + \langle x \rangle_{1}^{R} \langle y \rangle_{0} \\
\langle xy \rangle_{1}^{I} = \langle x \rangle_{0} \langle y \rangle_{1}^{I} + \langle x \rangle_{1}^{I} \langle y \rangle_{0}
\end{cases}$$
(12)

Finally, the Equation (12) is applied to the state equation of the DAB converter to establish its small-signal model.

#### 3.3. Discrete-Time Modeling

The discrete-time modeling method is also capable of capturing the high-frequency dynamic behavior of DAB and has been widely employed in power electronics modeling [52–55]. As demonstrated by the generalized averaging model approach discussed earlier, it is based on continuous-time function models. In contrast, for discrete-time modeling, the discrete state equations are obtained by sampling the DAB at different time instances, resulting in changes to the coefficient matrix of the differential equations in the state equation. In the case of discrete modeling for DAB under SPS modulation, four different switching states are utilized for discrete sampling, and the previous switching state is used to represent the next switching state. Through iterative iterations, the final discrete state equations are obtained, and by applying the Z-transform, the transfer function of the discrete system can be derived.

The continuous model of DAB can be obtained through the modeling of the state variables:

$$\dot{X} = AX + BU$$

$$Y = CX$$
(13)

Discrete-time model:

$$X(n) = FX(n-1) + GU$$

$$Y(n) = CX(n-1)$$
(14)

The coefficients of the state variables,  $A_1$ ,  $A_2$ ,  $A_3$ ,  $A_4$ , and  $B_1$ ,  $B_2$ ,  $B_3$ ,  $B_4$ , under different switching states can be obtained from the switching function.

The solution to the continuous state equation is generally expressed as:

$$X(t) = E(t - t_0)X(t_0) + \int_{t_0}^t E(t - \tau)BU(\tau)dt$$
(15)

where  $E(t) = e^{At}$ . Specifically for DAB, considering only one period of  $T_s$ , there are four switch states:

$$\begin{cases}
S_1: X_1 = f_1(X_4, \varphi) \Leftrightarrow X_1 = F(\varphi)X_4 + G(\varphi)U_1, t \in (0, t_1) \\
S_2: X_2 = f_2(X_1, \varphi) \Leftrightarrow X_2 = F(\varphi)X_1 + G(\varphi)U_1, t \in (t_1, t_2) \\
S_3: X_3 = f_3(X_2, \varphi) \Leftrightarrow X_3 = F(\varphi)X_2 + G(\varphi)U_1, t \in (t_2, t_3) \\
S_4: X_4 = f_4(X_3, \varphi) \Leftrightarrow X_4 = F(\varphi)X_3 + G(\varphi)U_1, t \in (t_3, t_4)
\end{cases}$$
(16)

where  $\varphi$  represents the phase shift angle,  $t_1 = \frac{\varphi}{2\pi}T_s$ ,  $t_2 = (1 - \frac{\varphi}{\pi})\frac{T_s}{2}$ ,  $t_3 = 2t_1 + t_2$ ,  $t_4 = 2t_1 + 2t_2$ .

Substituting Equation (16) into Equation (15) with DAB's four states as discrete intervals yields provides the following:

$$\begin{cases} X_{1} = f_{1}(x_{4}, \varphi) = e^{A_{1}t_{1}}X_{4} + \int_{0}^{t_{1}} e^{A_{1}t_{1}}B_{1}dtU_{1} = e^{A_{1}t_{1}}X_{4} + A_{1}^{-1}(e^{A_{1}t_{1}} - I)B_{1}U_{1} \\ X_{2} = f_{2}(x_{1}, \varphi) = e^{A_{2}t_{2}}X_{1} + \int_{t_{1}}^{t_{2}} e^{A_{2}t_{2}}B_{2}dtU_{1} = e^{A_{2}t_{2}}X_{4} + A_{2}^{-1}(e^{A_{2}t_{2}} - I)B_{2}U_{1} \\ X_{3} = f_{3}(x_{2}, \varphi) = e^{A_{3}t_{3}}X_{2} + \int_{t_{2}}^{t_{3}} e^{A_{3}t_{3}}B_{3}dtU_{1} = e^{A_{3}t_{3}}X_{4} + A_{3}^{-1}(e^{A_{3}t_{3}} - I)B_{3}U_{1} \\ X_{4} = f_{4}(x_{3}, \varphi) = e^{A_{4}t_{4}}X_{3} + \int_{t_{3}}^{t_{4}} e^{A_{4}t_{4}}B_{4}dtU_{1} = e^{A_{4}t_{4}}X_{4} + A_{4}^{-1}(e^{A_{4}t_{4}} - I)B_{4}U_{1} \end{cases}$$
(17)

By repeatedly iterating (17), a general discrete expression can be obtained:

$$X_{n+1} = f(x_n, \varphi) = f_1(f_4(f_3(f_2(x_1, \varphi), \varphi), \varphi), \varphi) = F(\varphi)X_n + G(\varphi_n)U_1$$
(18)

#### 3.4. General Modeling Based on Fourier Series

It can be seen from the previous three modeling methods that different modulation methods require the establishment of corresponding models. Advanced modulation methods offer a greater number of controllable variables. However, as the controllable variables of the high-frequency link increase, the system model changes with different time periods and operating states, making it quite complex to establish a model that can be uniformly described. References [56,57] provide a unified description of the DAB high-frequency link characteristics based on Fourier analysis and establish a unified model for DAB under different modulation methods.

The modulation techniques of SPS, DPS, and EPS can be seen as particular cases that fit into the wider framework of TPS modulation, and they can be represented in a unified form. The high-frequency link voltages exist in three different phase-shift variables:  $\beta_1$ ,  $\beta_2$ , and  $\alpha$ . Here,  $\beta_1$  and  $\beta_2$  are the internal phase angles of the primary and secondary sides, respectively, while  $\alpha$  represents the external phase shift between the primary and secondary sides. SPS, DPS, and EPS can all be considered as special cases of this form, where  $\beta_1 = \beta_2 = 0$  is applied for SPS modulation,  $\beta_1 = \beta_2$  is applied for DPS modulation, and  $\beta_1 = 0$  or  $\beta_2 = 0$  is applied for EPS modulation.

The primary and secondary side high-frequency link voltages can be expressed in Fourier series as:

$$v_{AB}(t) = \sum_{n=1,3,5,\dots} \frac{4V_{in}}{n\pi} \cos\left(\frac{n\beta_1}{2}\right) \sin(n\omega_0 t)$$

$$v_{CD}(t) = \sum_{n=1,3,5,\dots} \frac{4V_{out}}{n\pi} \cos\left(\frac{n\beta_2}{2}\right) \sin([n(\omega_0 t - \alpha)]n)$$
(19)

The equation for leakage inductance current can be derived by:

$$\begin{cases} i_s(t) - i_s(0) = \int_0^t \frac{v_{AB}(t) - v_{CD}(t)}{L_s} dt \\ i_s\left(\frac{\pi}{\omega_0}\right) = -i_s(0) \end{cases}$$
(20)

$$i_s(t) = \sum_{n=1,3,5} \frac{4}{n^2 \pi \omega_0 L_s} \sqrt{A^2 + B^2} \sin\left(n\omega_0 t + \arctan\frac{A}{B}\right)$$
(21)

$$\begin{cases} A = V_{out} \cos\left(n\frac{\beta_2}{2}\right) \cos(n\alpha) - V_{in} \cos\left(n\frac{\beta_1}{2}\right) \\ B = V_{out} \cos\left(n\frac{\beta_2}{2}\right) \sin(n\alpha) \end{cases}$$
(22)

Based on Equations (19) and (21), we can obtain a unified transfer power function for the high-frequency link:

$$P = \sum_{n=1,3,5,\dots} \frac{8V_{in}V_{out}}{n^3\pi^2\omega_0 L_s} \cos\left(n\frac{\beta_1}{2}\right) \cos\left(n\frac{\beta_2}{2}\right) \sin(n\alpha)$$
(23)

Equations (19), (21) and (23) provide the time-varying behaviors of voltage, current, and instantaneous power at each frequency component, which together form the high-frequency link voltage, current, and power. These results are consistent with the ones obtained from the reduced-order modeling analysis. Thus, the steady-state model can be used to derive the DAB small-signal model.

#### 3.5. Comparison of Modeling Methods

In this section, a comparison of different modeling methods is conducted based on their application scenarios and modeling accuracy.

Regarding the application scenarios, the reduced-order modeling method is employed in three areas. Firstly, it is used to analyze the steady-state characteristics of the DAB under different modulation methods [34,58,59]. The reduced-order modeling method effectively expresses the relationships between electrical quantities in the DAB circuit, enabling a detailed description of the operating states. Secondly, this method establishes mathematical models for optimization strategies with fewer variables [45,60,61]. Lastly, it is employed in the design of control methods based on Proportional–Integral (PI) controllers [62–67]. By developing a reduced-order model that expresses the average values of output power, output current, output voltage, and phase shift angle, small-signal models can be derived to establish transfer functions, facilitating controller design. The generalized state-space averaging method finds applications in advanced control methods. In [68,69], this method is employed to establish an adaptive model under TPS modulation, followed by the design of an adaptive controller. The generalized state-space averaging model provides information about the dynamic response characteristics of the DAB. By formulating the generalized state-space equations of the system and performing linearization, transfer functions or state-space models can be obtained. The model is useful for analyzing system frequency response, stability, and control design. Discrete-time modeling is primarily applied in DAB model predictive control. In [47,70], the discrete-time model is used to construct predictive models for predicting the future behavior of the system. By utilizing a discrete-time model of the DAB, in combination with current measurements and control inputs at the current time, one-step or multi-step predictions of the DAB operating states can be made. The general modeling method based on the Fourier series is mainly employed to establish mathematical models for optimization strategies with multiple variables [44,71,72].

Regarding modeling accuracy, reduced-order modeling is a technique to simplify complex system models. It achieves this by neglecting certain high-order dynamics or nonlinear characteristics of the system, reducing the model to a lower-order one. Although the DAB is simplified to a lower-order model, it does not compromise the accuracy of the model, as this is determined by the inherent characteristics of the DAB topology [73]. In the case of the generalized state-space averaging method, while a full-order model theoretically should achieve higher accuracy, the accuracy may be compromised when significant harmonic distortions are present since the full-order model is based on the first harmonic only [74]. Discrete-time modeling methods can provide higher modeling accuracy, particularly suitable for dynamic behavior analysis and control design of the system. The general modeling method based on the Fourier series is theoretically the most accurate and general modeling method. However, in practical systems, high-order effects are often neglected to trade off accuracy for practicality. The accuracy of the four modeling methods is further verified through simulation. The DAB simulation parameters are shown in Table 1.

Parameter	Value
C2	650 μF
$L_s$	84 μΗ
$R_{eq}$	7.2 Ω
n	7:5
$f_s$	10 kHz
$D_2$	0.2
$R_t$	0.01 Ω

Table 1. The DAB parameters for model accuracy comparison.

Figure 8 illustrates the comparative results between different modeling methods and the actual model. The accuracy of these methods in reflecting the output voltage  $V_{out}$  is evaluated by varying the input voltage  $V_{in}$ . At t = 0.04, the input voltage rises from 700 V to 1400 V, and at t = 0.08, it changes from 1400 V to 2100 V. From Figure 8, it can be observed that the reduced-order modeling method exhibits the highest precision with no steady-state error compared with the actual model. The discrete-time modeling method incurs a relatively small steady-state error. On the other hand, both the generalized state-space averaging and Fourier series-based generalized modeling methods display lower accuracy, primarily due to the consideration of only the fundamental harmonic component. These simulation results are in alignment with theoretical analyses.



Figure 8. Modeling method accuracy comparison.

In summary, the choice of modeling method depends on the specific application scenario and the desired modeling accuracy. Reduced-order modeling techniques are suitable for analyzing steady-state characteristics, establishing optimization models with fewer variables, and designing control methods. The generalized state-space averaging method is employed in advanced control techniques, providing insight into dynamic response characteristics. Discrete-time modeling is utilized in DAB model predictive control, enabling predictions of future system behavior. The general modeling method based on the Fourier series is applied to establish mathematical models for optimization strategies with multiple variables. The accuracy of each modeling method varies, with reduced-order modeling sacrificing some accuracy for simplicity, while discrete-time modeling provides higher accuracy, and the general modeling method based on the Fourier series offers the most accuracy in theory but may neglect higher-order effects for practicality.

#### 4. Optimization Algorithms

The high-efficiency operation of the DAB converter is a key technical challenge in power conversion and application. Optimization objectives such as minimizing inductor current stress, reducing losses, optimizing the circulating power, and expanding the softswitching range are commonly pursued in order to improve transmission efficiency and related performance. Through the combination of different modulation methods and solving algorithms, a variety of optimal modulation strategies are proposed. The traditional methods for solving the optimization problem of the DAB converter are mainly based on the Karush-Kuhn-Tucker (KKT) conditions using the Lagrange multiplier approach, later referred to as the KKT approach. In recent years, numerous intelligent algorithms, such as swarm intelligence optimization and reinforcement learning algorithms, have been proposed successively to solve DAB performance optimization problems under different modulation methods. These solution algorithms have opened up new technical avenues for DAB converter efficiency optimization and have become promising research directions. In the following sections, the current research status of the KKT approach, swarm intelligence optimization, and reinforcement learning algorithms in solving DAB modulation optimization problems will be reviewed and analyzed.

#### 4.1. KKT Approach

Initially, most optimization strategies for modulation in the DAB converter were primarily based on the Lagrange Multiplier Method (LMM) [61,75]. The KKT approach can be considered as an extension of the LMM. In the context of the optimization problem in DAB, the KKT algorithm constructs a Lagrangian function to convert the inequality constraints into equality constraints, thereby transforming the optimization problem into solving a set of nonlinear equations. By solving this set of nonlinear equations, the optimal solution and Lagrange multipliers can be obtained, enabling the derivation of the optimization modulation algorithm for the DAB can be described as follows:

N

$$\operatorname{Ain} J_n(Y) \tag{24}$$

$$P_n(\mathbf{Y}) - P_{on} = 0, I_k \leq 0, k = 1, \dots, q$$
 (25)

where  $Y = (D_1, D_2, D_3)$ , the objective function  $J_n$  represents different optimization objectives such as the current stress characteristic function, and circulating power characteristic function, among others. Here, n denotes the number of switching modes. The equality constraint is represented by  $P_n(Y) - P_{on}$ , while  $I_k$  represents the inequality constraints determined by operational constraints of effective switching modes. The number of inequality constraints is denoted by k. Mathematically, the local minimum point  $Y^*$  satisfies the KKT conditions, which can be described as follows:

$$L(\mathbf{Y}, \lambda, \mu) = J_n(\mathbf{Y}) + \lambda [P_n(\mathbf{Y}) - P_{on}] + \sum_{k=1}^{q} \mu_k I_k(\mathbf{Y})$$
  

$$\frac{\partial L}{\partial \mathbf{Y}}\Big|_{\mathbf{Y} = \mathbf{Y}^*} = 0, \lambda \neq 0$$
  

$$P_n(\mathbf{Y}^*) - P_{on} = 0$$
  

$$\mu_k \ge 0, \ I_k(\mathbf{Y}^*) \le 0$$
  

$$\mu_k B_k(\mathbf{Y}^*) = 0, k = 1, \dots, q$$
(26)

where *L* is the Lagrangian function, and  $\lambda$  as well as  $\mu_k$  are defined as the KKT multiplier.

In Reference [76], under the constraint of full-range soft-switching conditions, the KKT algorithm is employed to determine the phase-shift angles D1, D2, and D3 for the TPS modulation method, aiming to minimize the inductor current stress. Reference [44] proposes a variable duty cycle modulation approach to minimize the peak-to-peak current of the inductor for the DAB converter in four different operating modes. The KKT algorithm is applied to solve this optimization problem. Furthermore, Reference [60] presents an optimization scheme using EPS modulation to reduce the circulating power in DAB converters. The KKT method is employed to obtain the optimal solution.

The KKT method is proficient in handling convex optimization problems. However, it also possesses limitations. Specifically, when applied to DAB optimization, the KKT method requires separate optimization for different operating modes, thereby increasing the complexity of the solution process. For instance, in the case of optimizing the TPS modulation, which encompasses a total of 12 operating modes, the values need to be recalculated and compared for all 12 modes whenever circuit parameters change, resulting in significant computational time. DAB is frequently employed as an interface for new energy generation and energy storage systems, characterized by stochastic and fluctuating behaviors. This dynamic nature causes rapid variations in the voltages on both sides of the DAB, and the computational time required for optimization algorithms can significantly affect the effectiveness of DAB optimization and control. As a result, most optimizations are limited to offline calculations, followed by the use of look-up tables in microprocessors for optimized operation. Furthermore, the KKT method can only optimize a single objective and does not facilitate the simultaneous optimization of multiple objectives. Consequently, in recent years, researchers have applied advanced algorithms to optimize DAB modulation strategies, aiming to achieve the coordinated optimization of circulating power, soft-switching range, current stress, and other characteristics of the DAB system.

#### 4.2. Genetic Algorithm

The inception of the genetic algorithm (GA) can be attributed to its initial proposition by J. H. Holland in the year 1969. Genetic algorithms are a type of evolutionary algorithm that use principles inspired by natural selection to search for solutions to optimization problems. To apply a GA to the modulation problem in DAB converters, the problem must first be formulated as an optimization problem with a set of objective functions and constraints. The objective functions could include minimizing current stress, maximizing efficiency, or minimizing circulating power, depending on the specific requirements of the application. Once the optimization problem is formulated, the GA can be used to find the phase shift ratio that meets the desired objectives. The GA iteratively evaluates the fitness of each potential solution in the population, selecting the best solutions for reproduction and applying crossover and mutation operations to create new solutions.

For the multi-objective optimization modulation problem of DAB, it can be uniformly described as:

Obj.1 : 
$$J_1(V_{in}, V_{out}, D_1, D_2, D_3)$$
  
Obj.2 :  $J_2(V_{in}, V_{out}, D_1, D_2, D_3)$   
...  
Obj.n :  $J_n(V_{in}, V_{out}, D_1, D_2, D_3)$  (27)  
Equeonst.1 :  $P(V_{in}, V_{out}, D_1, D_2, D_3) = P_{on}$   
Inequeonst.1 : switching model $(D_1, D_2, D_3)$   
Inequeonst.2 :  $ZVS(V_{in}, V_{out}, D_1, D_2, D_3)$ .

The objective functions, Obj.*n*, in the context of DAB modulation, can represent characteristics such as current stress, circulating power, and RMS current of the inductor. The equality constraint, Equconst.1, is commonly used in DAB modulation to ensure that the actual transmitted power matches the desired transmitted power. The inequality constraint, Inequconst.1, represents the constraints under different switching models. Inequconst.2 represents the zero voltage switching (ZVS) constraint. Equation (27) plays a crucial role in constructing the fitness function of the genetic algorithm. It is used to evaluate and select the generated offspring, making it a critical step in the optimization process of modulation using genetic algorithms.

A multi-objective efficiency optimization scheme based on a genetic algorithm and TPS modulation is proposed in Reference [77]. This optimization scheme encompasses three objectives: minimizing RMS current, reducing current stress, and achieving ZVS performance. By employing a genetic algorithm, the proposed approach effectively tackles the non-convex optimization problem encountered in the multi-objective modulation of the DAB system. Reference [78] introduces a genetic algorithm-based droop control method aimed at reducing power losses in DAB converters and thereby enhancing transmission efficiency. Furthermore, Reference [79] presents the application of a genetic algorithm for multi-objective optimization modulation of modular DC-DC converters.

The genetic algorithm excels in addressing multi-objective cooperative optimization problems and non-convex optimization problems that cannot be directly solved by using the KKT approach. However, its search capability is limited, and it imposes high demands on parameter settings. When there are variations in circuit parameters, corresponding adjustments to algorithm parameters are required. Furthermore, in the online computation of optimized modulation for DAB converters, real-time and dynamic considerations are necessary. The search process of genetic algorithms typically involves a substantial number of iterations and fitness evaluations, which may lead to high computational complexity and difficulty in meeting real-time requirements.

#### 4.3. Particle Swarm Optimization Algorithm

The Particle Swarm Optimization (PSO) algorithm was initially proposed by J. Kennedy and R. Eberhart in 1995 [80]. Both PSO and genetic algorithms are regarded as swarm intelligence search algorithms, with their core lying in the construction of the fitness function. For the optimization modulation problem of DAB, the fitness function can be constructed based on Equation (27), and the specific construction method is elaborated upon in Reference [71]. Here, a concise overview of the fundamental concept of PSO is presented. In the PSO algorithm, each solution to be optimized can be conceptualized as a particle, and the motion state of a particle is represented by its position and velocity. The particle's position denotes the parameter values of the solution to be optimized, i.e.,  $x(D_1, D_2, D_3)$ , while its velocity, denoted as v, represents the search direction and step size of the particle in the solution space. The particle's motion state is influenced by the historical best position of the particle itself and the swarm's historical best position. Throughout the search process, particles continuously update their positions and velocities in order to discover more favorable solutions. The formulas for updating a particle's position and velocity are as follows:

$$v_{i,d}^{t+1} = \omega v_{i,d}^t + c_1 r_{1,d} \left( pbest_{i,d} - x_{i,d}^t \right) + c_2 r_{2,d} \left( gbest_d - x_{i,d}^t \right)$$
(28)

$$x_{i,d}^{t+1} = x_{i,d}^t + v_{i,d}^{t+1}$$
(29)

where,  $v_{i,d}^t$  represents the velocity of the *i*-th particle in the *d*-th dimension,  $x_{i,d}^t$  represents the position of the *i*-th particle in the *d*-th dimension,  $pbest_{i,d}$  represents the historical best position of the entire group,  $\omega$  is the inertia weight,  $c_1$  and  $c_2$  are the individual and social learning factors of the particle, and  $r_{1,d}$  and  $r_{2,d}$  are random numbers in the range of [0, 1]. By continuously updating the positions and velocities of particles, PSO can search for optimal solutions in the solution space.

A modulation strategy for optimizing the reflux power based on the TPS modulation method and a unified model using the Fourier series was proposed in Reference [71]. The PSO algorithm was utilized to solve for the phase-shift angle corresponding to the minimum circulating power. This method also increased the ZVS range in the DAB converter and reduced switching losses. Reference [79] presented a multivariable optimization modulation strategy solved using the PSO algorithm. This optimization modulation strategy included phase-shift modulation with PWM, effectively reducing the root mean square current of the DAB converter. In Reference [81], the PSO algorithm was employed to find the optimal phase-shift angle under the TPS modulation strategy to minimize the root mean square current across the entire operating range. However, this optimization scheme did not consider the ZVS performance of the DAB converter, resulting in significant switching losses. The PSO algorithm can be combined with other algorithms to overcome the problem of falling into local optima. Reference [82] proposed an EPS modulation strategy based on a combination of GA and PSO algorithms to optimize the circulating power of the DAB converter. Furthermore, Reference [83] utilized both the PSO algorithm and the LMM method to optimize the transmission efficiency of the DAB converter. The PSO algorithm was applied for local optimization, while the LMM method was used for global optimization.

Compared with GA, PSO exhibits faster convergence speed and lower parameter dependence. However, it is prone to becoming trapped in local optima and requires significant computational resources, particularly in high-dimensional and complex problems, which necessitates longer computation time and higher computational precision. The existing literature addresses this issue by employing offline calculations followed by table look-up methods for implementing optimization control in DAB systems.

# 4.4. Q-Learning Reinforcement Learning Algorithm

The Q-learning algorithm, as a classic method in reinforcement learning, is an approach that utilizes temporal differences to solve reinforcement learning control problems [84]. Its principles can be illustrated as shown in the diagram below.

Figure 9 illustrates the basic framework of the Q-learning reinforcement learning algorithm, which is a trial-and-error approach aimed at gradually improving the decision-making capabilities of the agent. During each training episode, the agent, based on the current state  $S_t$ , selects an action  $A_t$ . Subsequently, it receives a corresponding reward and transitions to the next state  $S_{t+1}$ . The objective of the agent is to discover the optimal policy in each discrete state to maximize the expected discounted cumulative reward. Through continuous interactions between the agent and the environment, an optimized policy corresponding to the maximum cumulative reward can be obtained. Q-learning employs the e-greedy method and the Bellman optimality equation to select new actions. Compared with other reinforcement learning algorithms, Q-learning features a simple Q-function that can be used online as the agent interacts with the environment. In the context of DAB optimization modulation, the Q-learning algorithm primarily aims to optimize the phase shifts  $D_1$ ,  $D_2$ , and  $D_3$  corresponding to various performance metrics. It maintains a Q-value table to record the learned experiences and retrieves the optimal phase shift ratios based on the indices of the Q-value table.



Figure 9. The Q-learning algorithm schematic diagram.

Reference [84] proposes an efficiency optimization modulation strategy based on a Q-learning algorithm and TPS. This method enhances the transmission efficiency of the DAB by optimizing the power loss rate under ZVS conditions. However, after the completion of Q-learning algorithm training, the corresponding optimization results need to be stored in a discrete lookup table, which makes continuous control inconvenient. To address this issue, Reference [85] introduces an RL+ANN-based TPS modulation strategy. Firstly, the Q-learning algorithm is employed to solve for the optimization strategy corresponding to the minimum current stress in the DAB, and the optimization results are stored in the Q-table. Subsequently, the ANN algorithm is applied to fit this Q-table. Based on this, the trained ANN agent functions as a fast proxy predictor, providing continuous optimization modulation strategies for the DAB across the entire operating range. However, this RL+ANN approach involves two training processes, which increases the computational complexity and requires more training time. Due to the limitations of traditional optimization-based modulation strategies, which rely on mathematical models of the DAB, they may not provide optimal performance under complex operating conditions. The Q-learning reinforcement learning algorithm can learn modulation strategies through trialand-error and automatic optimization, specifically tailored to the operating environment, thereby achieving optimal performance.

#### 4.5. Comparison and Application Suggestions of Different Advanced Algorithms

In this section, a comparison of the aforementioned advanced algorithms is conducted, and the following metrics are proposed: A. Algorithm application complexity, B. Optimization speed, C. Environmental adaptability, D. Multi-objective optimization capability, and E. Ability to achieve online continuous optimization. Subsequently, application recommendations are provided for different modulation methods. Table 2 compares various advanced algorithms and some combination algorithms based on the aforementioned metrics.

Algorithm	Reference	Algorithm Application Complexity	Optimization Speed	Environmental Adaptability	Multi- Objective Optimization Capability	Ability to Achieve Online Continuous Optimization
KKT	[44,62,86]	Low	Medium	Low	Low	NO
GA	[77–79]	Medium	Medium	Medium	Medium	NO
PSO	[58,59,81]	High	Slow	High	Medium	NO
Q-learning	[84]	High	Fast	High	High	NO
GA+PSO	[82]	High	Slow	Medium	High	NO
PSO+KKT	[83]	High	Slow	Medium	High	NO
QL+ANN	[85]	High	Fast	High	High	Yes

Table 2. Algorithm Comparison Table.

From Table 2, it can be observed that the KKT method is applicable to simple modulation techniques such as DPS and EPS, which have relatively straightforward circuit operating modes and enable the establishment of KKT optimization models. However, it may not be suitable for multi-objective optimization problems. The GA algorithm exhibits moderate performance across various modulation techniques and working environments; therefore, it can be employed for diverse applications in DAB. The PSO algorithm demonstrates excellent environmental adaptability, enabling it to adjust to changes in circuit parameters without frequent parameter tuning, unlike the GA algorithm. However, due to its tendency to converge to local optima, it is beneficial to combine PSO with other algorithms to address premature convergence issues. Promising approaches include the hybrid GA+PSO and PSO+KKT algorithms. Additionally, the Q-learning algorithm, based on reinforcement learning, exhibits robust environmental adaptivity and, when combined with ANN, can achieve online continuous optimization, providing it with a competitive edge.

# 5. Dynamic Control Method of DAB

In practical operation, the dynamic performance of a converter is also critical. When the operating conditions of the converter change, a fast dynamic response through the controller is required to bring the converter back to a stable state. A review and comparison of different control methods, including Feedback Control, Linearization Control, Feedforward Plus Feedback Control on Output Voltage, Disturbance-Observer-Based Control (DOBC), Feedforward Current Control (FFCC), and Model Predictive Current Control (MPC) is presented in Reference [75]. Reference [87] applied the generalized super-twisting algorithm to DAB control in aircraft battery systems. This algorithm ensures the robustness of control actions in the presence of current and voltage disturbances. The voltage tracking capability and load disturbance suppression performance were used as indicators for the various control methods. However, all the control methods were based on SPS modulation, and the combination of advanced modulation and control methods was not summarized. As discussed above, SPS modulation alone may not achieve efficient power transmission. Therefore, the combination of optimization modulation strategy and control methods is of great importance and is reviewed in this paper.

#### 5.1. Generic Voltage Feedback Control for DAB

Reference [88] proposes a control method called Generic Voltage Feedback Control (GVFC) that is applicable to different modulation strategies. The control structure under TPS modulation is shown in Figure 10, which has three control degrees of freedom and can be used for corresponding control of any modulation method. The proposed control scheme includes a single voltage control loop and a PCTL block, where PCTL stands for Peak Current Tracking and Limiting. The function of PCTL is to avoid magnetic core saturation mainly by limiting the peak current by stopping the phase angle shift outward. Initially, the optimal internal phase angle is determined based on specific performance criteria. Subsequently, the feedback voltage is compared with the reference voltage using a comparator, and the resulting error is transmitted to the voltage controller. The voltage controller output, along with the feedforward controller, is then used to set the external phase angle. Then, the peak current is limited through the PCTL block, achieving balanced control without current sensors.



Figure 10. Generic feedback voltage block diagram.

#### 5.2. Model Predictive Control for DAB

Utilizing model predictive control (MPC) with DAB converters primarily revolves around employing a mathematical model of the converter for forecasting the system's future dynamics and fine-tuning the control inputs to attain the desired performance objectives. In MPC, a finite-time horizon is defined over which the future behavior of the system is predicted, and a cost function is formulated to minimize the error between the predicted and desired outputs, as well as to satisfy any constraints on the control inputs and outputs. Once the model is established, a cost function is formulated that quantifies the performance objectives, such as minimizing the output voltage ripple, achieving a fast transient response, and maximizing efficiency. The cost function also takes into account any constraints on the control inputs and outputs, such as the maximum and minimum values of the switch duty ratio and the output voltage. Using the cost function and the model, an optimization problem is formulated and solved at each sampling instant to determine the optimal control inputs over the finite-time horizon [89,90]. The optimization problem can be solved using various techniques, such as quadratic programming or nonlinear programming.

The advantages of MPC applied to DAB converters include the ability to handle nonlinear dynamics and constraints, fast transient response, and the ability to predict and correct disturbances in real time. However, MPC requires significant computational resources and can suffer from numerical issues such as constraint violation or instability if the cost function or model is not well formulated [91].

MPC has been incorporated to tackle the optimization control challenges encountered in DAB converters, as discussed in References [70,92–94]. When it comes to achieving rapid dynamic responses, MPC emerges as a highly promising alternative to traditional power converter control techniques. Additionally, the TPS method stands out as a particularly effective modulation strategy for DAB converters. This approach mitigates circulating current, reduces current-related stress, minimizes EMI noise, and extends the zero-voltage switching (ZVS) range, as visually represented in Figure 11. Hence, in the existing literature, there has been the introduction of a TPS-based MPC approach as outlined in Reference [95]. This method incorporates a current stress optimization scheme based on TPS modulation to elevate dynamic performance while adhering to the minimal current stress constraint, ensuring the maintenance of the desired output voltage level. A predictive model has been meticulously devised to accurately forecast the dynamic behavior of the output voltage in response to variations in input voltage and load disturbances. Notably, the model development is rooted in TPS modulation, facilitating the computation of the internal phase shift of the H-bridge through the current stress optimization TPS scheme. Results from both simulations and experimental tests provide compelling evidence for the merits of this proposed control algorithm, characterized by rapid dynamic response, absence of output voltage overshoot, consistent switching frequency, low computational complexity, and robust stability.



Figure 11. Model Predictive Control based on TPS block diagram.

The input of the MPC controller is the sampled voltage of DAB at time k,  $V_{out}(k)$ , and the reference voltage  $V_{out}^*$ . Using  $V_{out}(k)$  as input, the future output voltage  $\hat{V}_{out}(k + j/k)$  at each future sampling time within the prediction horizon N is predicted, where j = 1...N. The specific calculation formula is as follows:

$$\hat{V}_{out}(k+1/k) = V_{out}(k) + \frac{V_{in}}{4Cf_s^2 L_s} \left(\alpha^2(k/k) + 2\beta_1\beta_2(k/k) + \beta_1\beta_2\right) - \frac{i_0}{f_s C}$$
(30)

As shown in Equation (30),  $\hat{V}_{out}(k + 1/k)$  also depends on the future control input signal  $\alpha_3(k + j/k)$ . To accurately and quickly track the desired output voltage, the control input signal is calculated by minimizing a cost function:

Cost function 
$$J = \sum_{j=1}^{N} \left[ \hat{V}_{out}(k+j/k) - V_{out}^{*}(k+j) \right]^{2}$$
 (31)

Subject to constraints:

$$\beta_1 - \beta_2 \leqslant \alpha \leqslant \min(\beta_1, 1 - \beta_2) \tag{32}$$

The control objective is to compute the sequence of future control signals  $\alpha(k)$ ,  $\alpha(k+1)$ ,  $\alpha(k+2)$ ..., which makes the output voltage  $\hat{V}_{out}(k+j/k)$  closest to the reference  $V_{out}^*(k+j)$ .

# 5.3. Model Reference Adaptive Control for DAB

Adaptive control theory is a powerful tool for controlling power electronic converters, which are used in a wide variety of applications ranging from renewable energy systems to electric vehicles [86]. The goal of adaptive control theory is to design control algorithms that can adapt to changes in the converter's operating conditions, such as changes in the load or input voltage. One of the key concepts in adaptive control theory is the use of adaptive algorithms to adjust the controller's parameters in response to changes in

the system's behavior. These algorithms are typically based on a mathematical model of the converter, which describes the relationship between the input and output voltages, currents, and other key variables. The adaptive algorithm uses this model to estimate the converter's parameters, such as the resistance and inductance of the load, and then uses this information to adjust the controller's parameters to achieve the desired performance. There are many different types of adaptive algorithms that can be used in power electronic converters, including model reference adaptive control (MRAC), self-tuning regulators (STR), and adaptive sliding mode control (ASMC) [86,96,97]. Each of these algorithms has its own strengths and weaknesses, and the choice of algorithm depends on the specific requirements of the application.

The advantages of adaptive control applied to DC-DC converters include improved performance in the presence of uncertainties, robustness to variations in the system parameters, and the ability to adapt to changes in the load and input voltage. However, adaptive control can be complex to implement, and the convergence of the adaptive algorithm may depend on the design of the adaptive law and the choice of the adaptation parameters.

References [68,97] proposed a control strategy for DAB converters based on the traditional MRAC with SPS modulation strategy. However, high-frequency oscillations are inevitable when choosing large adaptive gains to accelerate asymptotic tracking, as traditional MRAC only guarantees the asymptotic stability of the system without considering its dynamic performance. In pursuit of enhancing the dynamic performance of the DAB converter, Reference [69] introduces an upgraded model reference adaptive control (MRAC) strategy founded on TPS modulation. This strategy enhances the system's tracking capability and bolsters its dynamic response by introducing tracking errors into the reference model within the MRAC framework. Moreover, to comprehensively assess the influence of multidimensional modulation schemes on the MRAC model, a generalized averaging model (GAM) is established using the TPS modulation scheme as its basis, building upon the foundation of the uniform-phase-shift (UPS) modulation scheme. Ultimately, the efficacy of this proposed approach is confirmed through simulation experiments. The control method is shown in Figure 12.



Figure 12. The control block diagram of MRAC.

The reference model for DAB converters is formally established as follows:

$$\frac{d}{dt}V_{out}^{*}(t) = -a_m V_{out}^{*}(t) + b_m r(t) + c_m e(t)$$
(33)

where  $a_m$ ,  $b_m$ , and  $c_m$  are considered positive. Then, the control law is obtained in the form of state feedback in Equation (34).

$$u(t) = a_r(t)r(t) + a_x(t)x(t)$$
(34)

The adaptive rule governing the defined parameters is as follows:

$$\begin{cases} \frac{d}{dt}a_r(t) = -\alpha e_m r(t) \\ \frac{d}{dt}a_x(t) = -\alpha e_m x(t) \end{cases}$$
(35)

where  $e_m = x(t) - y_m(t)$  and  $\alpha$  is considered positive.

The error system of IMRAC is:

$$\dot{e}_m(t) = (a_m - c_m)e_m(t) + g\tilde{a}_x x(t) + g\tilde{a}_r r(t)$$
(36)

$$\begin{cases} \widetilde{a}_r = a_r - \frac{b_m}{g} \\ \widetilde{a}_x = a_x - \frac{a_p - a_m}{g} \end{cases}$$
(37)

The Lyapunov energy function is defined as:

$$V = \frac{1}{2}e_m^2 + \frac{1}{2\alpha}g\left(\tilde{a}_r^2 + \tilde{a}_x^2\right)$$
(38)

It can be demonstrated through a straightforward derivation that the Lyapunov energy function exhibits negative definiteness.

$$\frac{d}{dt}V = -a_m e_m^2 < 0 \tag{39}$$

#### 5.4. Active Disturbance Rejection Control for DAB

The main idea of Active Disturbance Rejection Control (ADRC) is to define an extended state variable, which includes both internal and external disturbances, and then use an extended state observer (ESO) to observe the total disturbance of the system and compensate for it in the controller. The input and output of the system are simplified to be transformed only by a pure integrator and can then be controlled by a PD controller [98,99].

The biggest advantage of ADRC is that it does not require an exact mathematical model of the controlled system, which gives it an engineering practicality that other intelligent control methods may lack. For the most practical engineering systems, it is difficult to establish an accurate mathematical model, and the system parameters may change with operating conditions, requiring the controller to have good robustness. The ADRC has received widespread attention in recent years because of its good robustness, low requirement for system models, and better dynamic performance compared with PI control. However, the ADRC method may require high computational resources for real-time implementation, and the design of the nonlinear state observer and the controller can be challenging.

References [100,101] applied the ADRC method to DAB dynamic control, and [102] proposed an artificial neural network-based ADRC to regulate constant output voltage quickly and accurately under different operating conditions. Reference [103] used a particle swarm optimization algorithm to tune the parameters of the ADRC. Currently, the design of ADRC for DAB is based on the reduced-order modeling method to establish a first-order equivalent model of the DAB, and then design a second-order ESO. The specific control method is shown in Figure 13.



Figure 13. The control block diagram of ADRC.

To design a first-order active disturbance rejection controller (ADRC), a first-order model of the system is established. Based on the reduced small-signal model of the DAB converter in Section 3.1, the time-domain expression of the DAB converter can be derived.

$$C_2 \frac{d\hat{v}_{\text{out}}}{dt} = \frac{nV_{in}(1-2D)}{2fL_s}\hat{d} - \frac{\hat{v}_{\text{out}}}{R_2}$$
(40)

The expression in Equation (40) in a general form is as follows:

$$\dot{y} = -a_1 y + w + bu \tag{41}$$

The control objective is to maintain a stable output voltage, with the control variable being the phase-shift ratio *D*. The output *y* represents the output voltage of the converter, while the controller output *u* represents the phase-shift ratio *D*. The variable *w* represents disturbances within and outside the system, while  $a_1$  is an internal parameter of the DAB converter that is unknown, as well as *w*. The input control gain *b* is partially known, with the known part denoted as  $b_0$ , where  $b_0 = nV_{in}(1-2D)/2fL_s$ . Therefore, the equation can be written as:

$$\dot{y} = -a_1y + w + (b - b_0u) + b_0u = f + b_0u \tag{42}$$

Equation (42) considers the uncertainties in the internal parameters of the system as well as the external disturbances. Although the inductance parameter ls is included in b0, the effect of inductance can be neglected due to the compensating action of the total disturbance.

By selecting the state variables  $x_1 = y$ ,  $x_2 = f$ , the state vector  $\mathbf{x} = [x_1x_2]^T$  includes both the system state variables and the total disturbance. Let h = f(y, w), then the system state equation can be obtained as follows:

$$\begin{cases} \dot{x}_1 = x_2 + b_0 u \\ \dot{x}_2 = h \\ y = x_1 \end{cases}$$
(43)

For a first-order system, it is not necessary to observe the differential signals of the system state variables. Therefore, the first-order linear error state feedback law can be expressed as:

$$u_0 = k_p (v - z_1) \tag{44}$$

where the term  $u_0$  represents the output of LSEF and  $k_p$  represents the gain of the proportional controller.

Traditional PI controllers can eliminate system errors through integration, but they may also reduce system stability. The first-order LADRC can use second-order LESO to compensate for generalized disturbances in real time, effectively avoiding the negative impacts caused by integration. By estimating the total disturbance of the system using expanded state variables and compensating for the input end of the system, the control system can achieve better performance.

#### 5.5. Comparison of Different Dynamic Control Methods

GVFC is a popular control method based on the Proportional–Integral (PI) control strategy. It offers a straightforward and easily implementable approach, delivering satisfactory steady-state performance under stable operating conditions. However, its effectiveness is limited when confronted with nonlinearities and uncertainties, leading to inadequate dynamic response, especially in scenarios characterized by rapidly changing operating conditions. Consequently, it is not suitable for applications demanding precise performance under varying operating conditions.

MPC exhibits favorable tracking performance and disturbance rejection capabilities for DAB in various operating modes. Since DAB converters can establish accurate mathematical models, MPC can effectively handle nonlinearities and uncertainties. However, MPC's computational complexity and resource requirements are significant, necessitating powerful hardware support. Additionally, if the operating environment of the DAB converter system undergoes changes, it may lead to model variations that undermine the effectiveness of model predictive control.

In comparison to the aforementioned methods, MRAC overcomes its limitations by adapting to system uncertainties, variations, and nonlinearities without relying on precise system models. It can deliver good tracking and disturbance rejection capabilities under dynamically changing operating conditions. However, its implementation may be complex, requiring extensive system identification and parameter estimation processes.

ADRC exhibits resilience against disturbances and uncertainties. It effectively handles nonlinearities and time-varying dynamics without relying on precise system models, ensuring satisfactory tracking performance and disturbance rejection. However, the design of its core component, the ESO, is challenging, as poor design choices can hinder effective tracking of state variables and subsequently impact control performance.

In conclusion, while the generalized voltage feedback control is a simple and widely used strategy with good steady-state performance under stable operating conditions, it is limited in handling nonlinearities and uncertainties, making it unsuitable for applications with stringent performance requirements or varying operating conditions. MPC provides good tracking and disturbance rejection capabilities for DAB in different operating modes, but its implementation complexity and reliance on accurate system models may pose challenges. MRAC and ADRC offer solutions to these limitations, adapting to uncertainties, variations, and nonlinearities without requiring precise system models. However, they may involve complex implementation procedures and design considerations.

# 6. Conclusions

This passage presents a comprehensive review of various aspects concerning the research on DAB, encompassing modulation strategies, optimization algorithms, modeling methods, and advanced control techniques. The findings of DAB research are summarized as follows:

Two categories of modulation strategies are comprehensively introduced: PSM strategy and ADM strategy. The PSM strategy encompasses SPS, DPS, EPS, and TPS modulation methods, albeit essentially summarized by the TPS modulation method. The ADM strategy further extends the PSM strategy by introducing control over the conduction time of the same bridge arm switches. Continuing with the analysis, a comparative evaluation of various modulation methods' complexities and flexibilities was conducted, alongside a simulation-based comparison of the power characteristics of these methods. Recently, scholars have proposed ADM strategies, which offer greater flexibility compared with traditional PSM techniques. ADM modulation has emerged as one of the recent focal points in DAB research.

The commonly employed modeling methods for DAB were reviewed, including reduced-order modeling, generalized state-space averaging modeling, discrete-time modeling, and Fourier series-based universal modeling. Each modeling method is applicable to different scenarios and with varying levels of precision. Reduced-order modeling is widely adopted due to its simplicity and broad applicability. Generalized state-space averaging modeling offers higher modeling accuracy while simplifying circuit state analysis by reducing the required variables and equations. However, its modeling accuracy is compromised by neglect of higher-order harmonics in practical applications, and its complex theoretical foundation lacks the clear physical concepts found in reduced-order modeling. Discrete-time modeling is primarily used to incorporate high-frequency dynamics, particularly in advanced control methods. Fourier series-based universal modeling presents a comprehensive approach to establishing DAB operation models under various modulations, streamlining the analysis process at the expense of increased computational complexity. The diverse optimization algorithms are subsequently presented, including the KKT approach, GA, PSO, and Q-learning algorithm. The KKT approach is suitable for simple modulation techniques but not well-suited for multi-objective optimization. The GA exhibits moderate performance across different modulation techniques. The PSO algorithm demonstrates excellent environmental adaptability but may suffer from premature convergence. Combining PSO with other algorithms such as GA or KKT can address this issue. The Q-learning algorithm, when combined with artificial neural networks, exhibits robust environmental adaptivity and capability to achieve online continuous optimization, making it a promising approach.

The advanced dynamic control methods for DAB are reviewed: GVFC, MPC, MRAC, and ADRC. When GVFC is straightforward, it has limitations in handling non-linearity and uncertainties. MPC offers good performance but requires powerful hardware and accurate DAB models. MRAC and Active ADRC adapt to uncertainties without relying on DAB models, albeit posing challenges in implementation and design due to their complexity.

With the introduction of advanced modulation strategies, the modeling and control methods of DAB face new challenges. Despite some existing literature that combines advanced control methods with advanced optimization modulation methods, such studies remain relatively scarce, and some control methods have yet to incorporate advanced modulation strategies. Moreover, most articles simplify DAB to a first-order model for controller design, which compromises the controller's performance under significant disturbances. Furthermore, the advanced control methods are predominantly investigated under the framework of phase-shift modulation strategies, lacking research inegration with asymmetric PWM strategies, which represents a notable research gap.

**Author Contributions:** Every author played a role in enhancing the manuscript's quality. To elaborate, J.H. was responsible for conceiving the research idea, conducting theoretical analyses, and carrying out experiments. Y.C. and J.L. contributed by designing the methodology and experiments, as well as participating in the paper's writing and revision. J.C. and L.C. provided valuable input concerning the paper's structure, language, and format. Y.W. also contributed to the writing and revision of the paper. All authors have read and agreed to the published version of the manuscript.

**Funding:** This work was supported by the National Natural Science Foundation of China under grant No. 62273104 and the Science and Technology Planning Project of Guangdong Province under grant No. 2021A1515012398.

**Data Availability Statement:** No new data were created or analyzed in this study. Data sharing is not applicable to this article.

Acknowledgments: This work was supported by the National Natural Science Foundation of China under grant No. 62273104 and the Science and Technology Planning Project of Guangdong Province under grant No. 2021A1515012398.

Conflicts of Interest: The authors declare no conflict of interest.

# References

- 1. Dudley, B. BP Statistical Review of World Energy 2022; British Petroleum Co: London, UK, 2022.
- Li, B.; Lee, F.C.; Li, Q.; Liu, Z. Bi-Directional on-Board Charger Architecture and Control for Achieving Ultra-High Efficiency with Wide Battery Voltage Range. In Proceedings of the 2017 IEEE Applied Power Electronics Conference and Exposition (APEC), Tampa, FL, USA, 26–30 March 2017; pp. 3688–3694.
- Xu, Y.; Singh, C. Power System Reliability Impact of Energy Storage Integration with Intelligent Operation Strategy. In Proceedings of the 2014 IEEE PES General Meeting | Conference & Exposition, Dallas, TX, USA, 27–31 July 2014; p. 1.
- Burkart, R.M.; Kolar, J.W. Comparative η– ρ– σ Pareto Optimization of Si and SiC Multilevel Dual-Active-Bridge Topologies With Wide Input Voltage Range. *IEEE Trans. Power Electron.* 2017, *32*, 5258–5270. [CrossRef]
- 5. Saafan, A.A.; Khadkikar, V.; Moursi, M.S.E.; Zeineldin, H.H. A New Multiport DC-DC Converter for DC Microgrid Applications. *IEEE Trans. Ind. Appl.* **2023**, *59*, 601–611. [CrossRef]
- Dragičević, T.; Lu, X.; Vasquez, J.C.; Guerrero, J.M. DC Microgrids—Part I: A Review of Control Strategies and Stabilization Techniques. *IEEE Trans. Power Electron.* 2016, 31, 4876–4891.
- Wen, H.; Xiao, W.; Su, B. Nonactive Power Loss Minimization in a Bidirectional Isolated DC–DC Converter for Distributed Power Systems. *IEEE Trans. Ind. Electron.* 2014, 61, 6822–6831. [CrossRef]
- 8. Nuutinen, P.; Pinomaa, A.; Peltoniemi, P.; Kaipia, T.; Karppanen, J.; Silventoinen, P. Common-Mode and RF EMI in a Low-Voltage DC Distribution Network with a PWM Grid-Tie Rectifying Converter. *IEEE Trans. Smart Grid* **2017**, *8*, 400–408. [CrossRef]
- 9. Suryanarayana, H.; Sudhoff, S.D. Design Paradigm for Power Electronics-Based DC Distribution Systems. *IEEE J. Emerg. Sel. Top. Power Electron.* **2017**, *5*, 51–63. [CrossRef]
- 10. Cuzner, R.M.; Singh, V. Future Shipboard MVdc System Protection Requirements and Solid-State Protective Device Topological Tradeoffs. *IEEE J. Emerg. Sel. Top. Power Electron.* **2017**, *5*, 244–259. [CrossRef]
- 11. Wu, T.-F.; Kuo, C.-L.; Lin, L.-C.; Chen, Y.-K. DC-Bus Voltage Regulation for a DC Distribution System with a Single-Phase Bidirectional Inverter. *IEEE J. Emerg. Sel. Top. Power Electron.* **2016**, *4*, 210–220. [CrossRef]
- 12. Páez, J.D.; Frey, D.; Maneiro, J.; Bacha, S.; Dworakowski, P. Overview of DC–DC Converters Dedicated to HVdc Grids. *IEEE Trans. Power Deliv.* 2019, 34, 119–128. [CrossRef]
- 13. De Doncker, R.W.A.A.; Divan, D.M.; Kheraluwala, M.H. A Three-Phase Soft-Switched High-Power-Density DC/DC Converter for High-Power Applications. *IEEE Trans. Ind. Appl.* **1991**, *27*, 63–73. [CrossRef]
- 14. Ramachandran, R.; Nymand, M. Experimental Demonstration of a 98.8% Efficient Isolated DC–DC GaN Converter. *IEEE Trans. Ind. Electron.* **2017**, *64*, 9104–9113. [CrossRef]
- Zhao, B.; Song, Q.; Li, J.; Xu, X.; Liu, W. Comparative Analysis of Multilevel-High-Frequency-Link and Multilevel-DC-Link DC–DC Transformers Based on MMC and Dual-Active Bridge for MVDC Application. *IEEE Trans. Power Electron.* 2018, 33, 2035–2049. [CrossRef]
- Zhao, B.; Song, Q.; Li, J.; Sun, Q.; Liu, W. Full-Process Operation, Control, and Experiments of Modular High-Frequency-Link DC Transformer Based on Dual Active Bridge for Flexible MVDC Distribution: A Practical Tutorial. *IEEE Trans. Power Electron.* 2017, 32, 6751–6766. [CrossRef]
- 17. Ortiz, G.; Leibl, M.G.; Huber, J.E.; Kolar, J.W. Design and Experimental Testing of a Resonant DC–DC Converter for Solid-State Transformers. *IEEE Trans. Power Electron.* 2017, *32*, 7534–7542. [CrossRef]
- Tan, N.M.L.; Abe, T.; Akagi, H. Design and Performance of a Bidirectional Isolated DC–DC Converter for a Battery Energy Storage System. *IEEE Trans. Power Electron.* 2012, 27, 1237–1248. [CrossRef]
- 19. Karthikeyan, V.; Gupta, R. Multiple-Input Configuration of Isolated Bidirectional DC–DC Converter for Power Flow Control in Combinational Battery Storage. *IEEE Trans. Ind. Inform.* **2018**, *14*, 2–11. [CrossRef]
- She, X.; Huang, A.Q.; Ni, X. Current Sensorless Power Balance Strategy for DC/DC Converters in a Cascaded Multilevel Converter Based Solid State Transformer. *IEEE Trans. Power Electron.* 2014, 29, 17–22. [CrossRef]
- Shi, J.; Gou, W.; Yuan, H.; Zhao, T.; Huang, A.Q. Research on Voltage and Power Balance Control for Cascaded Modular Solid-State Transformer. *IEEE Trans. Power Electron.* 2011, 26, 1154–1166. [CrossRef]
- 22. Huber, J.E.; Kolar, J.W. Applicability of Solid-State Transformers in Today's and Future Distribution Grids. *IEEE Trans. Smart Grid* 2019, *10*, 317–326. [CrossRef]

- 23. Agamy, M.S.; Dong, D.; Garcés, L.J.; Zhang, Y.; Dame, M.E.; Wu, X.; Pan, Y. A High Power Medium Voltage Resonant Dual Active Bridge for MVDC Ship Power Networks. *IEEE J. Emerg. Sel. Top. Power Electron.* **2017**, *5*, 88–99. [CrossRef]
- 24. Mohammadpour, A.; Parsa, L.; Todorovic, M.H.; Lai, R.; Datta, R.; Garces, L. Series-Input Parallel-Output Modular-Phase DC–DC Converter With Soft-Switching and High-Frequency Isolation. *IEEE Trans. Power Electron.* **2016**, *31*, 111–119. [CrossRef]
- 25. Rahman, M.I.; Ahmed, K.H.; Jovcic, D. Analysis of DC Fault for Dual-Active Bridge DC/DC Converter Including Prototype Verification. *IEEE J. Emerg. Sel. Top. Power Electron.* **2019**, *7*, 1107–1115. [CrossRef]
- Xue, L.; Shen, Z.; Boroyevich, D.; Mattavelli, P.; Diaz, D. Dual Active Bridge-Based Battery Charger for Plug-in Hybrid Electric Vehicle With Charging Current Containing Low Frequency Ripple. *IEEE Trans. Power Electron.* 2015, 30, 7299–7307. [CrossRef]
- Everts, J.; Krismer, F.; Van den Keybus, J.; Driesen, J.; Kolar, J.W. Charge-Based ZVS Soft Switching Analysis of a Single-Stage Dual Active Bridge AC-DC Converter. In Proceedings of the 2013 IEEE Energy Conversion Congress and Exposition, Denver, CO, USA, 15–19 September 2013; pp. 4820–4829.
- 28. Krismer, F.; Kolar, J.W. Accurate Power Loss Model Derivation of a High-Current Dual Active Bridge Converter for an Automotive Application. *IEEE Trans. Ind. Electron.* 2010, *57*, 881–891. [CrossRef]
- Taylor, A.; Liu, G.; Bai, H.; Brown, A.; Johnson, P.M.; McAmmond, M. Multiple-Phase-Shift Control for a Dual Active Bridge to Secure Zero-Voltage Switching and Enhance Light-Load Performance. *IEEE Trans. Power Electron.* 2018, 33, 4584–4588. [CrossRef]
- Inoue, S.; Akagi, H. A Bidirectional Isolated DC–DC Converter as a Core Circuit of the Next-Generation Medium-Voltage Power Conversion System. *IEEE Trans. Power Electron.* 2007, 22, 535–542. [CrossRef]
- Xie, Y.; Sun, J.; Freudenberg, J.S. Power Flow Characterization of a Bidirectional Galvanically Isolated High-Power DC/DC Converter Over a Wide Operating Range. *IEEE Trans. Power Electron.* 2010, 25, 54–66.
- Oggier, G.G.; Garcla, G.O.; Oliva, A.R. Switching Control Strategy to Minimize Dual Active Bridge Converter Losses. *IEEE Trans.* Power Electron. 2009, 24, 1826–1838. [CrossRef]
- Krismer, F.; Kolar, J.W. Closed Form Solution for Minimum Conduction Loss Modulation of DAB Converters. *IEEE Trans. Power Electron.* 2012, 27, 174–188. [CrossRef]
- Zhao, B.; Yu, Q.; Sun, W. Extended-Phase-Shift Control of Isolated Bidirectional DC–DC Converter for Power Distribution in Microgrid. *IEEE Trans. Power Electron.* 2012, 27, 4667–4680. [CrossRef]
- Bai, H.; Nie, Z.; Mi, C.C. Experimental Comparison of Traditional Phase-Shift, Dual-Phase-Shift, and Model-Based Control of Isolated Bidirectional DC–DC Converters. *IEEE Trans. Power Electron.* 2010, 25, 1444–1449.
- Naayagi, R.T.; Forsyth, A.J.; Shuttleworth, R. Performance Analysis of Extended Phase-Shift Control of DAB DC-DC Converter for Aerospace Energy Storage System. In Proceedings of the 2015 IEEE 11th International Conference on Power Electronics and Drive Systems, Sydney, Australia, 9–15 June 2015; pp. 514–517.
- 37. Choi, W.; Rho, K.-M.; Cho, B.-H. Fundamental Duty Modulation of Dual-Active-Bridge Converter for Wide-Range Operation. *IEEE Trans. Power Electron.* **2016**, *31*, 4048–4064. [CrossRef]
- 38. Krismer, F.; Kolar, J.W. Accurate Small-Signal Model for the Digital Control of an Automotive Bidirectional Dual Active Bridge. *IEEE Trans. Power Electron.* **2009**, *24*, 2756–2768. [CrossRef]
- 39. Muthuraj, S.S.; Kanakesh, V.K.; Das, P.; Panda, S.K. Triple Phase Shift Control of an LLL Tank Based Bidirectional Dual Active Bridge Converter. *IEEE Trans. Power Electron.* **2017**, *32*, 8035–8053. [CrossRef]
- Tong, A.; Hang, L.; Li, G.; Jiang, X.; Gao, S. Modeling and Analysis of a Dual-Active-Bridge-Isolated Bidirectional DC/DC Converter to Minimize RMS Current with Whole Operating Range. *IEEE Trans. Power Electron.* 2018, 33, 5302–5316. [CrossRef]
- 41. Wu, K.; de Silva, C.W.; Dunford, W.G. Stability Analysis of Isolated Bidirectional Dual Active Full-Bridge DC–DC Converter with Triple Phase-Shift Control. *IEEE Trans. Power Electron.* **2012**, 27, 2007–2017. [CrossRef]
- 42. Hu, S.; Li, X.; Bhat, A.K.S. Operation of a Bidirectional Series-Resonant Converter with Minimized Tank Current and Wide ZVS Range. *IEEE Trans. Power Electron.* **2019**, *34*, 904–915. [CrossRef]
- Lu, M.; Hu, S.; Tang, Y.; Zhou, S.-Z.; Li, X. A Novel Control Strategy Based on Modified Gating Scheme for a Dual-Active-Bridge Converter. In Proceedings of the 2017 12th IEEE Conference on Industrial Electronics and Applications (ICIEA), Siem Reap, Cambodia, 18–20 June 2017; pp. 1576–1581.
- 44. Mou, D.; Luo, Q.; Wang, Z.; Li, J.; Wei, Y.; Shi, H.; Du, X. Optimal Asymmetric Duty Modulation to Minimize Inductor Peak-to-Peak Current for Dual Active Bridge DC–DC Converter. *IEEE Trans. Power Electron.* 2021, *36*, 4572–4584. [CrossRef]
- 45. Wang, M.; Wei, S.; Mou, D.; Wu, P. Research on Efficient Single-Sided Asymmetric Modulation Strategy for Dual Active Bridge Converters in Wide Voltage Range. *IEEE J. Emerg. Sel. Top. Power Electron.* **2022**, *10*, 5738–5748. [CrossRef]
- Rodríguez Alonso, A.R.; Sebastian, J.G.; Lamar, D.; Hernando, M.M.; Vazquez, A. An Overall Study of a Dual Active Bridge for Bidirectional DC/DC Conversion. In Proceedings of the 2010 IEEE Energy Conversion Congress and Exposition, Atlanta, GA, USA, 12–16 September 2010; pp. 1129–1135.
- 47. Chen, L.; Lin, L.; Shao, S.; Gao, F.; Wang, Z.; Wheeler, P.W.; Dragičević, T. Moving Discretized Control Set Model-Predictive Control for Dual-Active Bridge with the Triple-Phase Shift. *IEEE Trans. Power Electron.* **2020**, *35*, 8624–8637. [CrossRef]
- Bai, H.; Mi, C.; Wang, C.; Gargies, S. The Dynamic Model and Hybrid Phase-Shift Control of a Dual-Active-Bridge Converter. In Proceedings of the 2008 34th Annual Conference of IEEE Industrial Electronics, Orlando, FL, USA, 10–13 November 2008; pp. 2840–2845.
- 49. Sanders, S.R.; Noworolski, J.M.; Liu, X.Z.; Verghese, G.C. Generalized Averaging Method for Power Conversion Circuits. *IEEE Trans. Power Electron.* **1991**, *6*, 251–259. [CrossRef]

- 50. Caliskan, V.A.; Verghese, O.C.; Stankovic, A.M. Multifrequency Averaging of DC/DC Converters. *IEEE Trans. Power Electron.* **1999**, *14*, 124–133. [CrossRef]
- 51. Qin, H.; Kimball, J.W. Generalized Average Modeling of Dual Active Bridge DC–DC Converter. *IEEE Trans. Power Electron.* 2012, 27, 2078–2084.
- Costinett, D.; Zane, R.; Maksimović, D. Discrete-Time Small-Signal Modeling of a 1 MHz Efficiency-Optimized Dual Active Bridge Converter with Varying Load. In Proceedings of the 2012 IEEE 13th Workshop on Control and Modeling for Power Electronics (COMPEL), Kyoto, Japan, 10–13 June 2012; pp. 1–7.
- 53. Shi, L.; Lei, W.; Li, Z.; Huang, J.; Cui, Y.; Wang, Y. Bilinear Discrete-Time Modeling and Stability Analysis of the Digitally Controlled Dual Active Bridge Converter. *IEEE Trans. Power Electron.* 2017, *32*, 8787–8799. [CrossRef]
- 54. Verghese, G.C.; Elbuluk, M.E.; Kassakian, J.G. A General Approach to Sampled-Data Modeling for Power Electronic Circuits. *IEEE Trans. Power Electron.* **1986**, *PE-1*, 76–89. [CrossRef]
- Zhao, C.; Round, S.D.; Kolar, J.W. Full-Order Averaging Modelling of Zero-Voltage-Switching Phase-Shift Bidirectional DC–DC Converters. *IET Power Electron.* 2010, 3, 400–410. [CrossRef]
- Zhao, B.; Song, Q.; Liu, W.; Liu, G.; Zhao, Y. Universal High-Frequency-Link Characterization and Practical Fundamental-Optimal Strategy for Dual-Active-Bridge DC-DC Converter Under PWM Plus Phase-Shift Control. *IEEE Trans. Power Electron.* 2015, 30, 6488–6494. [CrossRef]
- Rahman, M.I.; Jovcic, D.; Ahmed, K.H. Generalised Fourier Series Model for Dual Active Bridge DC/DC Converter Based on Triple Phase Shift Modulation Method. In Proceedings of the 2021 21st International Symposium on Power Electronics (Ee), Novi Sad, Serbia, 27–30 October 2021; pp. 1–6.
- 58. Bai, H.; Mi, C. Eliminate Reactive Power and Increase System Efficiency of Isolated Bidirectional Dual-Active-Bridge DC–DC Converters Using Novel Dual-Phase-Shift Control. *IEEE Trans. Power Electron.* **2008**, *23*, 2905–2914. [CrossRef]
- 59. Zhao, B.; Song, Q.; Liu, W. Power Characterization of Isolated Bidirectional Dual-Active-Bridge DC–DC Converter with Dual-Phase-Shift Control. *IEEE Trans. Power Electron.* **2012**, 27, 4172–4176. [CrossRef]
- 60. Shi, H.; Wen, H.; Chen, J.; Hu, Y.; Jiang, L.; Chen, G.; Ma, J. Minimum-Backflow-Power Scheme of DAB-Based Solid-State Transformer with Extended-Phase-Shift Control. *IEEE Trans. Ind. Appl.* **2018**, *54*, 3483–3496. [CrossRef]
- 61. Zhao, B.; Song, Q.; Liu, W. Efficiency Characterization and Optimization of Isolated Bidirectional DC–DC Converter Based on Dual-Phase-Shift Control for DC Distribution Application. *IEEE Trans. Power Electron.* **2013**, *28*, 1711–1727. [CrossRef]
- 62. Zhao, W.; Zhang, X.; Gao, S.; Ma, M. Improved Model-Based Phase-Shift Control for Fast Dynamic Response of Dual-Active-Bridge DC/DC Converters. *IEEE J. Emerg. Sel. Top. Power Electron.* **2021**, *9*, 223–231. [CrossRef]
- 63. Song, W.; Hou, N.; Wu, M. Virtual Direct Power Control Scheme of Dual Active Bridge DC–DC Converters for Fast Dynamic Response. *IEEE Trans. Power Electron.* 2018, 33, 1750–1759. [CrossRef]
- Qin, H.; Kimball, J.W. Closed-Loop Control of DC–DC Dual-Active-Bridge Converters Driving Single-Phase Inverters. *IEEE Trans.* Power Electron. 2014, 29, 1006–1017.
- 65. Tong, A.; Hang, L.; Chung, H.S.-H.; Li, G. Using Sampled-Data Modeling Method to Derive Equivalent Circuit and Linearized Control Method for Dual-Active-Bridge Converter. *IEEE J. Emerg. Sel. Top. Power Electron.* **2021**, *9*, 1361–1374. [CrossRef]
- 66. Shan, Z.; Jatskevich, J.; Iu, H.H.-C.; Fernando, T. Simplified Load-Feedforward Control Design for Dual-Active-Bridge Converters with Current-Mode Modulation. *IEEE J. Emerg. Sel. Top. Power Electron.* **2018**, *6*, 2073–2085. [CrossRef]
- 67. Dutta, S.; Hazra, S.; Bhattacharya, S. A Digital Predictive Current-Mode Controller for a Single-Phase High-Frequency Transformer-Isolated Dual-Active Bridge DC-to-DC Converter. *IEEE Trans. Ind. Electron.* **2016**, *63*, 5943–5952. [CrossRef]
- Brando, G.; Del Pizzo, A.; Meo, S. Model-Reference Adaptive Control of a Dual Active Bridge Dc-Dc Converter for Aircraft Applications. In Proceedings of the 2018 International Symposium on Power Electronics, Electrical Drives, Automation and Motion (SPEEDAM), Amalfi, Italy, 20–22 June 2018; pp. 502–506.
- Yang, H.; Yang, L. Model Reference Adaptive Control Strategy for Dual Active Bridge Converter with Improved Reference Model. In Proceedings of the 2022 International Conference on Power Energy Systems and Applications (ICoPESA), Singapore, 25–27 February 2022; pp. 298–303.
- Hagawane, T.; Sindhu, M.R.; Patankar, R. Model Predictive Control of Dual Active Bridge DC-DC Converter Based on PWM plus TPS Modulation for Automotive Applications. In Proceedings of the 2022 International Virtual Conference on Power Engineering Computing and Control: Developments in Electric Vehicles and Energy Sector for Sustainable Future (PECCON), Chennai, India, 5–6 May 2022; pp. 1–6.
- 71. Shi, H.; Wen, H.; Hu, Y.; Jiang, L. Reactive Power Minimization in Bidirectional DC–DC Converters Using a Unified-Phasor-Based Particle Swarm Optimization. *IEEE Trans. Power Electron.* **2018**, *33*, 10990–11006. [CrossRef]
- 72. Li, J.; Luo, Q.; Mou, D.; Wei, Y.; Sun, P.; Du, X. A Hybrid Five-Variable Modulation Scheme for Dual-Active-Bridge Converter with Minimal RMS Current. *IEEE Trans. Ind. Electron.* **2022**, *69*, 336–346. [CrossRef]
- 73. Shao, S.; Chen, L.; Shan, Z.; Gao, F.; Chen, H.; Sha, D.; Dragicevic, T. Modeling and Advanced Control of Dual-Active-Bridge DC–DC Converters: A Review. *IEEE Trans. Power Electron.* **2022**, *37*, 1524–1547. [CrossRef]
- Wang, P.; Liu, C.; Guo, L. Modeling and Simulation of Full-Bridge Series Resonant Converter Based on Generalized State Space Averaging. AMM 2013, 347–350, 1828–1832. [CrossRef]
- Hou, N.; Song, W.; Wu, M. Minimum-Current-Stress Scheme of Dual Active Bridge DC–DC Converter with Unified Phase-Shift Control. *IEEE Trans. Power Electron.* 2016, 31, 8552–8561. [CrossRef]

- 76. Huang, J.; Wang, Y.; Li, Z.; Lei, W. Unified Triple-Phase-Shift Control to Minimize Current Stress and Achieve Full Soft-Switching of Isolated Bidirectional DC–DC Converter. *IEEE Trans. Ind. Electron.* **2016**, *63*, 4169–4179. [CrossRef]
- 77. Zhou, L.; Gao, Y.; Ma, H.; Krein, P.T. Wide-Load Range Multiobjective Efficiency Optimization Produces Closed-Form Control Solutions for Dual Active Bridge Converter. *IEEE Trans. Power Electron.* **2021**, *36*, 8612–8616. [CrossRef]
- 78. Meng, L.; Dragicevic, T.; Vasquez, J.C.; Guerrero, J.M. Tertiary and Secondary Control Levels for Efficiency Optimization and System Damping in Droop Controlled DC–DC Converters. *IEEE Trans. Smart Grid* **2015**, *6*, 2615–2626. [CrossRef]
- Chen, P.; Liu, J.; Xiao, F.; Zhu, Z.; Ren, Q. Multi-Objective Optimization Design of Modular Multilevel Multi-Port DC-DC Converter Assessing Efficiency, Volume and Reliability. In Proceedings of the 2020 IEEE 9th International Power Electronics and Motion Control Conference (IPEMC2020-ECCE Asia), Nanjing, China, 29 November–2 December 2020; pp. 2749–2756.
- 80. Kennedy, J.; Eberhart, R. Particle Swarm Optimization. In Proceedings of the ICNN'95—International Conference on Neural Networks, Perth, WA, Australia, 27 November–1 December 1995; Volume 4, pp. 1942–1948.
- 81. Hebala, O.M.; Aboushady, A.A.; Ahmed, K.H.; Abdelsalam, I. Generic Closed-Loop Controller for Power Regulation in Dual Active Bridge DC–DC Converter with Current Stress Minimization. *IEEE Trans. Ind. Electron.* **2019**, *66*, 4468–4478. [CrossRef]
- Akboy, E.; Vural, R.A. Minimizing Reactive Power of DAB Converters in EPS Control with Evolutionary Computation. In Proceedings of the 2021 10th International Conference on Renewable Energy Research and Application (ICRERA), Istanbul, Turkey, 26–29 September 2021; pp. 351–356.
- Jean-Pierre, G.; Altin, N.; El Shafei, A.; Nasiri, A. Efficiency Optimization of Dual Active Bridge DC– DC Converter with Triple Phase-Shift Control. In Proceedings of the 2020 IEEE Energy Conversion Congress and Exposition (ECCE), Online, 11–15 October 2020; pp. 1217–1222.
- Tang, Y.; Hu, W.; Xiao, J.; Chen, Z.; Huang, Q.; Chen, Z.; Blaabjerg, F. Reinforcement Learning Based Efficiency Optimization Scheme for the DAB DC–DC Converter with Triple-Phase-Shift Modulation. *IEEE Trans. Ind. Electron.* 2021, 68, 7350–7361. [CrossRef]
- Tang, Y.; Hu, W.; Xiao, J.; Lu, Z.; Li, Z.; Huang, Q.; Chen, Z.; Blaabjerg, F. RL-ANN-Based Minimum-Current-Stress Scheme for the Dual-Active-Bridge Converter with Triple-Phase-Shift Control. *IEEE J. Emerg. Sel. Top. Power Electron.* 2022, 10, 673–689. [CrossRef]
- 86. Segaran, D.; McGrath, B.P.; Holmes, D.G. Adaptive Dynamic Control of a Bi-Directional DC-DC Converter. In Proceedings of the 2010 IEEE Energy Conversion Congress and Exposition, San Jose, CA, USA, 20–24 September 2010; pp. 1442–1449.
- 87. Russo, A.; Canciello, G.; Cavallo, A. Generalized Super-Twisting Control of a Dual Active Bridge for More Electric Aircraft. In Proceedings of the 2021 European Control Conference (ECC), Online, 29 June–2 July 2021; pp. 1610–1615.
- Yadav, Y.S.; Nayak, G.; Dasgupta, A. A Generic Current Sensorless Control Scheme for Dual Active Bridge Converter. In Proceedings of the 2022 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), Jaipur, India, 14–17 December 2022; pp. 1–6.
- 89. Xie, Y.; Ghaemi, R.; Sun, J.; Freudenberg, J.S. Model Predictive Control for a Full Bridge DC/DC Converter. *IEEE Trans. Control. Syst. Technol.* **2012**, *20*, 164–172. [CrossRef]
- 90. Zheng, L.; Kandula, R.P.; Kandasamy, K.; Divan, D. Stacked Low-Inertia Converter or Solid-State Transformer: Modeling and Model Predictive Priority-Shifting Control for Voltage Balance. *IEEE Trans. Power Electron.* **2021**, *36*, 8934–8952. [CrossRef]
- 91. Andrés-Martínez, O.; Flores-Tlacuahuac, A.; Ruiz-Martinez, O.F.; Mayo-Maldonado, J.C. Nonlinear Model Predictive Stabilization of DC–DC Boost Converters with Constant Power Loads. *IEEE J. Emerg. Sel. Top. Power Electron.* **2021**, *9*, 822–830. [CrossRef]
- 92. Cortes, P.; Rodriguez, J.; Quevedo, D.E.; Silva, C. Predictive Current Control Strategy with Imposed Load Current Spectrum. *IEEE Trans. Power Electron.* **2008**, *23*, 612–618. [CrossRef]
- Papafotiou, G.; Geyer, T.; Morari, M. Hybrid Modelling and Optimal Control of Switch-Mode Dc-Dc Converters. In Proceedings of the 2004 IEEE Workshop on Computers in Power Electronics, Urbana, IL, USA, 15–18 August 2004; pp. 148–155.
- Mariethoz, S.; Beccuti, A.G.; Papafotiou, G.; Morari, M. Sensorless Explicit Model Predictive Control of the DC-DC Buck Converter with Inductor Current Limitation. In Proceedings of the 2008 Twenty-Third Annual IEEE Applied Power Electronics Conference and Exposition, Austin, TX, USA, 24–28 February 2008; pp. 1710–1715.
- 95. Akbar, S.M.; Hasan, A.; Watson, A.J.; Wheeler, P. Model Predictive Control with Triple Phase Shift Modulation for a Dual Active Bridge DC-DC Converter. *IEEE Access* 2021, *9*, 98603–98614. [CrossRef]
- Soejima, T.; Ishizuka, Y.; Domoto, K.; Hirose, T. Adaptive Control Technique for High Power Efficiency Dual Active Bridge DC-DC Converter with Wide Load Range. In Proceedings of the 2018 IEEE Energy Conversion Congress and Exposition (ECCE), Portland, OR, USA, 23–27 September 2018; pp. 2829–2834.
- Mitra, S.K.; Karanki, S.B. A Grid Integrated Bi-Directional Dual Active Bridge Converter with Model Reference Adaptive Control Based Power Flow Controller. In Proceedings of the 2019 National Power Electronics Conference (NPEC), Tiruchirappalli, India, 13–15 December 2019; pp. 1–5.
- 98. Zhou, X.; Liu, Q.; Ma, Y.; Xie, B. DC-Link Voltage Research of Photovoltaic Grid-Connected Inverter Using Improved Active Disturbance Rejection Control. *IEEE Access* 2021, *9*, 9884–9894. [CrossRef]
- 99. Du, Y.; Cao, W.; She, J. Analysis and Design of Active Disturbance Rejection Control with an Improved Extended State Observer for Systems with Measurement Noise. *IEEE Trans. Ind. Electron.* **2023**, *70*, 855–865. [CrossRef]
- 100. Ali, M.; Yaqoob, M.; Cao, L.; Loo, K.H. Disturbance-Observer-Based DC-Bus Voltage Control for Ripple Mitigation and Improved Dynamic Response in Two-Stage Single-Phase Inverter System. *IEEE Trans. Ind. Electron.* **2018**, *66*, 6836–6845. [CrossRef]

- Wu, Y.; Mahmud, M.H.; Alhosaini, W.; Zhao, Y.; Mantooth, A.; Zhang, Y. An Uncertainty and Disturbance Estimator Based Voltage Control for Dual-Active-Bridge Converters. In Proceedings of the 2019 IEEE Energy Conversion Congress and Exposition (ECCE), 29 September–3 October 2019; pp. 5204–5209.
- Zeng, Y.; Maswood, A.I.; Pou, J.; Zhang, X.; Li, Z.; Sun, C.; Mukherjee, S.; Gupta, A.K.; Dong, J. Active Disturbance Rejection Control Using Artificial Neural Network for Dual-Active-Bridge-Based Energy Storage System. *IEEE J. Emerg. Sel. Top. Power Electron.* 2023, 11, 301–311. [CrossRef]
- Zeng, Y.; Zhang, X.; Mukherjee, S.; Gupta, A.K.; Sun, C.; Dong, J. Adaptive Active Disturbance Rejection Control of DAB Based on PSO. In Proceedings of the IECON 2020 The 46th Annual Conference of the IEEE Industrial Electronics Society, Singapore, 18–21 October 2020; pp. 2840–2845.

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.